Reviewer #1:
Scientific Quality: Grade A (Excellent)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (High priority)
Specific Comments to Authors: It is a very interesting research effort on a usual uroepithelial cancer of the bladder. Genomic factors analysis and targeted therapy in cancer pathways inhibiting carcinogenesis may be an effective prospect. The manuscript is well-written and documented with high-quality illustrations. Congratulations.

Answering review:
We would like to thank the reviewer for the affirmation of our manuscript. If we are fortunate enough to be accepted and published by the journal, we will continue our research in the field of cancer, focusing on genomic factor analysis and targeted therapy, in order to identify effective targets for cancer treatment and bring hope to more cancer patients.